Author's response to reviews

Title: A study protocol for applying the Co-creating Knowledge Translation Framework to a population health study

Authors:

Kathryn Powell (kathryn.powell@adelaide.edu.au)
Alison L Kitson (alison.kitson@adelaide.edu.au)
Elizabeth Hoon (elizabeth.hoon@adelaide.edu.au)
Jonathan Newbury (jonathan.newbury@adelaide.edu.au)
Anne Wilson (annewilson@gmail.com)
justin beilby (justin.beilby@adelaide.edu.au)

Version: 2 Date: 28 June 2013

Author's response to reviews:

letter from NHMRC on funding